

GOPEN ACCESS

Citation: Salloum T, Moussa R, Rahy R, Al Deek J, Khalifeh I, El Hajj R, et al. (2020) Expanded genome-wide comparisons give novel insights into population structure and genetic heterogeneity of *Leishmania tropica* complex. PLoS Negl Trop Dis 14(9): e0008684. https://doi.org/10.1371/journal. pntd.0008684

Editor: Yara M. Traub-Csekö, Instituto Oswaldo Cruz, BRAZIL

Received: January 30, 2020

Accepted: August 6, 2020

Published: September 18, 2020

Copyright: © 2020 Salloum et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: This Whole Genome Shotgun project has been deposited in DDBJ/ENA/ GenBank under the following accession numbers: LT1 (strain designation: MHOM/LB/2017/IK) NCBI BioProject accession: PRJNA438080 and GenBank: QBEU00000000.1; LT2 (strain designation: MHOM/LB/2015/IK) NCBI BioProject accession: PRJNA453461 and GenBank: QEH000000000.1; ATCC50129 (strain designation: MHOM/SU/74/ K27) NCBI BioProject accession: PRJNA589179 and GenBank: WLYJ0000000.1. RESEARCH ARTICLE

Expanded genome-wide comparisons give novel insights into population structure and genetic heterogeneity of *Leishmania tropica* complex

Tamara Salloum¹, Rim Moussa¹, Ryan Rahy¹, Jospin Al Deek¹, Ibrahim Khalifeh², Rana El Hajj², Neil Hall³, Robert P. Hirt⁴*, Sima Tokajian¹*

 Department of Natural Sciences, School of Arts and Sciences, Lebanese American University, Byblos, Lebanon, 2 Department of Pathology and Laboratory Medicine, American University of Beirut, Beirut, Lebanon, 3 Earlham Institute, Norwich research Park, University of East Anglia, Norwich, United Kingdom,
Biosciences Institute, Faculty of Medical Sciences, Newcastle University, Newcastle upon Tyne, United Kingdom

* robert.hirt@ncl.ac.uk (RPH); stokajian@lau.edu.lb (ST)

Abstract

Leishmania tropica is one of the main causative agents of cutaneous leishmaniasis (CL). Population structures of L. tropica appear to be genetically highly diverse. However, the relationship between L. tropica strains genomic diversity, protein coding gene evolution and biogeography are still poorly understood. In this study, we sequenced the genomes of three new clinical L. tropica isolates, two derived from a recent outbreak of CL in camps hosting Syrian refugees in Lebanon and one historical isolate from Azerbaijan to further refine comparative genome analyses. In silico multilocus microsatellite typing (MLMT) was performed to integrate the current diversity of genome sequence data in the wider available MLMT genetic population framework. Single nucleotide polymorphism (SNPs), gene copy number variations (CNVs) and chromosome ploidy were investigated across the available 18 L. tropica genomes with a main focus on protein coding genes. MLMT divided the strains in three populations that broadly correlated with their geographical distribution but not populations defined by SNPs. Unique SNPs profiles divided the 18 strains into five populations based on principal component analysis. Gene ontology enrichment analysis of the protein coding genes with population specific SNPs profiles revealed various biological processes, including iron acquisition, sterols synthesis and drug resistance. This study further highlights the complex links between L. tropica important genomic heterogeneity and the parasite broad geographic distribution. Unique sequence features in protein coding genes identified in distinct populations reveal potential novel markers that could be exploited for the development of more accurate typing schemes to further improve our knowledge of the evolution and epidemiology of the parasite as well as highlighting protein variants of potential functional importance underlying L. tropica specific biology.

Funding: This work was financed by the School of Arts and Sciences at the Lebanese American University, Office of Graduate Studies and Research (Grant number GSR-R0063) and Newcastle University that partially supported TS with an Overseas Research Scholarship. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Human cutaneous leishmaniasis (CL) is a parasitic infection transmitted through the bite of an infected sandfly vector. Members of the *Leishmania tropica* complex are one of the main causative agents of CL in the Old World. Despite having varying phenotypical manifestations and degrees of severity, the genomic differences underlying the observed phenotypic diversity are not well understood. CL outbreaks in Lebanon, Syria and in Azerbaijan and neighbouring countries are evidence that leishmaniasis is still an important public health concern. Here, we explored unique genetic features in L. tropica genomes by comparing the gene variants and gene copy number variations, as well as chromosome ploidy, of two recent isolates collected during a CL outbreak in Lebanon and one historical isolate from Azerbaijan with 16 published L. tropica genomes. In silico MLMT was also performed to integrate the current diversity of genome sequence data in the wider geography-related genetic populations. This study further highlights the distinct population structure and genomic heterogeneity of the L. tropica species complex in the context of the parasite broad geographic distribution. Unique sequence features identified in distinct populations could be exploited for the development of more accurate typing schemes and highlight proteins of potential functional importance to gain new insights into L. tropica specific biology.

Introduction

Leishmania species are sandfly vector-borne parasitic kinetoplastid protozoan that cause leishmaniasis, one of the major neglected tropical diseases [1–2]. Leishmaniasis has three major forms of pathologies including (i) cutaneous leishmaniasis (CL) appearing as self-healing skin ulcers or nodules, (ii) mucocutaneous leishmaniasis (MCL) that can lead to disfiguring mutilations and the destruction of the nasal and/or oral mucosa and (iii) systemic forms, visceral leishmaniasis (VL) also called kala-azar, which includes primarily gross inflammatory reactions within spleen and liver typically fatal if left untreated [1–3]. *Leishmania tropica* is a particularly heterogenous species complex displaying varying serological [4], biochemical [5] and genetic properties [6] and a broad geographical distribution across Africa and Eurasia [7]. Its clinical spectrum of diseases ranges across CL with varying lesions severity to VL [8] and includes some patients that have displayed post kala-azar dermal leishmaniasis after treatment [9], and occasional cases of nasopharyngeal MCL [10]. *L. tropica* can be transmitted by at least four sandfly vectors including *Phlebotomus sergenti*, *P. arabicus*, *P. guggisbergi* and *P. chabaudi* and a few additional species, which are thought to mediate anthroponotic, and potentially zoonotic, infections [2,11,12].

Microsatellite markers are commonly used for sub-species analysis of leishmanial populations and 21 such independents genetic markers specific for *L. tropica* identified 10 distinct genetic clusters [13]. A broader sampling of *L. tropica* strains analysed with a subset of 12 of these microsatellite markers identified three major distinct biogeographic populations: India/ Asia, Israel/Palestine and Africa/Galilee [14,15].

Genome-wide genotyping of *Leishmania spp*. allows to investigate genetic diversity at multiple levels including single nucleotide polymorphism (SNPs), copy number variations (CNVs), gene family complement and how these might relate to biogeography and their biology more generally [16]. However, additional genome sequences are needed from a range of *L. tropica* isolates covering various mammalian hosts, sandfly vectors and geographic origins to allow more in-depth comparative analyses to study in more detail the

evolutionary dynamics of this species. Among the 17 *Leishmania* annotated genomes available in the TriTrypDB database [17], currently only one belongs to the *L. tropica* species, strain LRC-L590, which was isolated from a human patient in Israel [4]. Recently, Iantorno and colleagues generated whole genome sequencing (WGS) data for 14 *L. tropica* isolates [18] and Bussotti and colleagues sequenced an additional *L. tropica* genome that wasn't compared to other *L. tropica* genomes [19]. In contrast, multiples genomes for other species of *Leishmania* are deposited to date on TriTrypDB and derived from diverse isolates/strains conferring VL (*L. infantum* and *L. donovani*) [20, 21], CL (*L. major, L. mexicana*) [22, 23] or MCL (*L. braziliensis*) [20]. Broad WGS samplings are available for some species, for examples comparative genomics analyses for 151 *L. donovani* strains were recently described [24].

Leishmania species have similar genomic arrangement consisting of 34 to 36 chromosomes (size range ~0.25 to ~4 megabases—Mb) [25]. Old World *Leishmania* spp. including *L. major* and *L. tropica* have 36 chromosomes [25], which are mostly diploid among the *L. tropica* strains with WGS data [18]. Aneuploidy can be observed among *Leishmania* genomes with varying degrees as are smaller amplified DNA segments, referred to as copy number variants (CNVs), associated with drug selective pressure or nutritional stress [26]. In general, the gene order and sequence are highly conserved among the 30 *Leishmania* species [26] despite their (i) high degree of variability in the symptomatology and the severity of the diseases caused by different species, (ii) their broad host range (both among mammals and sandflies vectors) and (iii) broad geographic distribution [20].

Following the crisis in the Syrian Arab Republic that started in March 2011, CL outbreaks caused by L. tropica have been reported in a number of Lebanese regions [27, 28]. In this study, we perform a comparative genome analysis of three novel L. tropica genomes: two recent (2014) clinical isolates representing the outbreak of CL in Lebanon and one historical isolate from Azerbaijan (1974), with previously generated genome sequence data from 16 isolates [18,19]. Notably, we integrated the 19 WGS data with a wide microsatellite based genotyping survey across over 160 L. tropica strains [14,15,29] contributing to rationalise comparative analyses of SNPs and CNVs as well as chromosome ploidy. This study provides new insights into the population structure and genomic diversity of L. tropica parasites and integrate these into a more global comparative framework. Notably, sampling of new strains from the Middle East is also of particular interest for comparative studies as this is one of the regions with the highest prevalence for L. tropica infections, which are characterised by a complex and broad genetic diversity apparently derived from distinct lineages [14,15,29]. Furthermore, CL appears to be re-emerging in various countries in the Middle East and Transcaucasia [27,30]. Hence, WGS data generated from isolates collected from these regions is also important for the study of the epidemiology of these outbreaks. Enriching the genome sequence data from these two regions also allows to study the nature of the links between L. tropica lineages from Africa versus those from East Asia [13,15,16], with the former potentially derived from the latter, with some authors suggesting an African origin for the Eurasian L. tropica complex [12, 31]. Identification of unique genomic features could also contribute to the development of more accurate genotyping schemes. These data will also help to direct future sampling of strains for strategic genome sequencing to eventually cover more effectively the tremendous genetic diversity of L. tropica across its broad geographic distribution. Genome wide comparative studies also provide important opportunities to identify sequence variations among protein coding genes (e.g. [21, 24] for L. donovani) that potentially underpin L. tropica specific biology including specific adaptations to accommodate variations across its broad biogeography and ecology [2, 7,11].

Methods

Ethical approval

The approval from the Institution Review Board (IRB) at American University of Beirut Medical Centre (AUBMC) was granted on August 1st 2011 prior to sample collection (approval reference #PALM I.K.01). All patients completed a risk assessment form. Recruitment in refugee's camps was through announcement of the pathologists visit through Doctors without Borders (Médecins Sans Frontières: https://www.msf.org). All samples derived from patients were treated anonymously and no personal details were recorded or used for this project, only the sample itself and the characteristics of the pathologies associated with the infection were considered. The obtained ethical approval was also processed and accepted as part of TS's PhD project approval at Newcastle University (Ref: 9663/2016).

Sample information and patient's metadata

A total of 19 *L. tropica* genome sequences, from strains collected between 1974 and 2016, were used for the below analyses, which included three newly generated genome sequence data. All isolates originated from humans CL cases except for LRC-L810 that was isolated from the sandfly *P. arabicus*. LT1 (strain designation: MHOM/LB/2017/IK, NCBI BioProject accession PRJNA438080; GenBank: QBEU00000000.1) and LT2 (strain designation: MHOM/LB/2015/ IK, NCBI BioProject accession PRJNA453461; GenBank: QEHO00000000.1) were isolated in 2014 from skin punch biopsies collected at the American University of Beirut Medical Centre (AUBMC) with detailed patients' annotations. Genomic DNA from isolate ATCC 50129 (thereafter ATCC50129) (strain designation: MHOM/SU/74/K27) (NCBI BioProject accession: PRJNA589179; GenBank: WLYJ0000000.1) was purchased from ATCC-LGC Partnership (Teddington Middlesex, UK). ATCC50129 was isolated in 1974 from a human host in Baku, Azerbaijan, which was at the time part of the former Union of Soviet Socialist Republics.

The genome of the reference strain LRC-L590 was isolated in 1990 from an 11-year-old male in Kfar Adumim, Judean Desert, Israel [4] and was analysed and re-sequenced by Iantorno and colleagues [18]. The 14 WGS data obtained from Iantorno and colleagues [18] were derived from Afghanistan, India, Israel, Jordan, Saudi Arabia and Syria (Table 1). Isolate LRC-L810 was collected from an infected sandfly, *P. arabicus*, from the Galilee region in northern Israel, and it was reported that this strain was preferentially transmitted by the *P. arabicus* vector rather than *P. sergenti* [18, 32]. The WGS data for Ltr16 was generated by Bussotti and colleagues [19]. It was collected from a human CL case in 2016 and obtained from the Pasteur Institute in Morocco. Information for all 19 strains/isolates considered here for comparative genomics analyses are summarized in S1 Table.

Histopathology & parasite culture

The tissue samples of LT1 and LT2 were stained for hematoxylin and eosin, Giemsa, Acid Fast bacilli, Gomori Methylamine Silver and Periodic Acid-Schiff and classified according to the modified Ridley's parasitic index (PI) as previously described [33]. The two samples had a PI = 5 indicating that \geq 10,000 amastigotes were present per standard tissue section. Skin punch biopsies were processed as previously described by El Hajj and colleagues [34]. *Leishmania* promastigotes were maintained in standard RPMI-1640 medium (Sigma # R8758) supplemented with 20% heat-inactivated fetal bovine serum (FBS) (Thermo # 10500064) and 1% penicillin-streptomycin (Lonza # 17-602E) with a temperature ranging from 22°C to 25°C. Stationary phase promastigotes were concentrated by centrifugation at 1,000 g for 8 min (Multifuge X1R, Thermo Fisher Scientific, Heraeus Holding GmbH, Germany) and washed twice

Table 1. Heterozygosity and homozygosity among a total of 560,735 polymorphic sites identified by analysing the genomes of 18 *L. tropica* isolates. The number of heterozygous (He) and homozygous (Ho) positions were calculated using as reference the isolate L590. *: Isolates sequenced in this study; ^a: Genome generated by Bussotti et al (2018) [19]; ^b: Genomes generated by Iantorno et al (2017) [18]. MLMT-based populations are added next to isolates' names as follows: A/G for Africa/Galilee, A/I for Asia/India and I/P for Israel/Palestine.

Isolate	Но	He	Total	Origin	PCA Population	MLMT
Ltr16 ^a	179,694	23,379	203,073	Morocco	Ι	A/G
MA-37 ^b	169,996	29,323	199,319	Jordan		
MN-11 ^b	177,160	32,302	209,462	Jordan		
Average	175,617	28,335	203,951			
E50 ^b	299,104	39,603	338,707	Israel	II	I/P
LRC-L747 ^b	300,336	40,532	340,868	Israel		
LRC-L810 ^b	280,556	39,535	320,091	Israel		A/G
Average	293,332	39,890	333,222			
Boone ^b	278,592	35,139	313,731	Saudi Arabia	III	A/G
Melloy ^b	211,103	94,470	305,573	Saudi Arabia		
Ackerman ^b	283,121	36,172	319,293	Israel		
K26_1 ^b	283,120	36,172	319,292	India		A/I
Average	263,984	50,488	314,472			
K112_1 ^b	307,639	38,998	346,637	India	IV	A/I
Azad ^b	303,378	37,321	340,699	Afghanistan		A/G
KK27 ^b	306,045	38,019	344,064	Afghanistan		
LT1*	309,690	41,482	351,172	Lebanon		
LT2*	275,046	22,810	297,856	Lebanon		
Rupert ^b	306,308	38,188	344,496	Afghanistan		
Kubba ^b	302,208	37,461	339,669	Syria		
Average	301,473	36,326	337,799			
ATCC50129*	162,528	38,598	201,126	Azerbaijan	V	A/I

https://doi.org/10.1371/journal.pntd.0008684.t001

with sterile phosphate-buffered saline solution (Sigma # D8662) to remove any FBS traces. Parasites were resuspended and counted using a Neubauer chamber (Haemocytometer) slide.

DNA extraction

For LT1 and LT2, DNA extraction from stationary growth phase promastigotes ($\sim 10^7$ cells) was performed using the Qiagen QIAamp DNA Mini Kit (Catalog #51304) according to the manufacturer's instructions. 200 µL of DNA was eluted in AE (10 mM Tris·Cl; 0.5 mM EDTA; pH 9.0) buffer. The extracted DNA had a concentration of \sim 30.7 ng/µL as measured on Qubit fluorometer 2.0 (Invitrogen, Carlsbad, CA, USA) and a purity (A260/A280) of 2.1.

Illumina sequencing and reads assembly

For LT1 and LT2 library construction and paired-end library sequencing was performed commercially (Outsourced through GECKO, Dubai, UAE). Briefly, sequencing for the two Lebanese isolates was performed using the Illumina HiSeq4000 platform and the TruSeq DNA PCR-Free library preparation kit according to the manufacturer's standard sequencing protocol. 100 bp reads were generated and checked for low sequence quality and presence of adapters using FastQC [35]. Trimmomatic was used to trim the reads with Phred score below 20 and remove adapters with a sliding window of 4:15 [36]. *L. tropica* ATCC50129 was sequenced by the Earlham Institute genomic pipelines group. TruSeq PCR free libraries were generated using the manufacturers' protocol and were sequenced in the Illumina Miseq Platform using the 250bp paired-end V2 chemistry.

Reads mapping and variant calling

Bowtie2 software [37] was used to map the sequence reads against the reference genome *L. tropica* L590 obtained from the TriTrypDB database (version 44) [17] (http://tritrypdb.org/ tritrypdb/). Samtools software were used to convert the resulting SAM files to BAM files before sorting them and marking the duplicates [38]. Variant calling was then performed using Freebayes software [39]. Tabix software was used to index the VCF files [40]. Individual VCF files were merged using bcftools (version 1.4) software for subsequent analysis (https://samtools. github.io/bcftools).

SNP analyses

A pairwise comparison of all the isolates' SNPs was then performed to determine the level of similarity between the SNPs detected in each isolate. The metric used to measure this similarity is the ratio of the number of common SNPs over the total number of SNPs between every two isolates. In other words, for every two isolates, the number of SNPs in their intersection was divided by the number of SNPs in their union. The number of common SNPs and total number of SNPs between every two isolates were obtained using the vcf-compare module of the vcftools (version 0.1.13) software [41]. A heatmap representing the results of this comparison was then generated using the heatmap.2 function from the gplots package in R (http://cran.r-project.org/web/packages/gplots/index.html).

Next, in order to visualise population structure based on SNPs profiles principal component analysis (PCA) was performed on the merged VCF file using the SNPRelate R package (version 1.16.0) [42] available through Bioconductor (version 3.9) software. A plot of the resulting principal component (PC) scores was generated using the ggplot2 R package (version 3.0.0). The grouping of the isolates in this plot was used to split the isolates into five different populations having <0.1 PC score differences. PCA was repeated on populations III, IV and V separately showing that ATCC50129 clustered further apart from K26_1 and thus was considered as a separate population.

Per isolate SNP heterozygosity was calculated from the same merged VCF file using the *het* argument of vcftools. In order to annotate the variants, the SnpEff tool was used [43]. The annotated L590 genome obtained from TriTrypDB (version 44) was used as a reference. Genetic variant annotation and functional effect prediction were extracted from the output of SnpEff software [43] and their distribution across the 18 isolates was summarized and tabulated (S2 Table).

Chromosome aneuploidy analysis

Prediction of chromosome aneuploidy was performed as previously described by Zackay and colleagues [44] and implemented in R. The algorithm relies on each chromosome's median read coverage, which was computed for each sorted BAM file using bedtools2 genome perbase reports. The median depth of known stable diploid chromosomes (chromosomes 34 & 36) was computed and considered the mean read depth for all diploid chromosomes as previously described [18]. The expected chromosome ploidy was calculated by finding the fold change in read depth with respect to the mean diploid read depth, and the results were saved in an 18x36 matrix (19 strains minus reference strain L590, 36 chromosomes). A visual representation of the chromosome ploidy was generated using the heatmap.2 function in the R package gtools (version 3.0.1.1). A heatmap showing the aneuploidy profiles of each isolate and a dendrogram clustering the isolates by Euclidean distances based on the similarity of the aneuploidy profiles were generated. The tool heatmap.2 reorders the dendrogram based on the row and column mean values.

Gene copy number variations

In order to detect the copy number variations (CNVs) of the different genes, the cn.mops (Copy Number estimation by a Mixture Of PoissonS) R package (version 1.28.0) was used from R-Bioconductor [45]. cn.MOPS provides integer copy numbers, estimates variations in read counts across isolates and uses these estimates for CNV calling. The algorithm implemented in this package uses BAM files to obtain read counts for each read, which are then normalized by the total number of reads. A sliding window approach of 1000 base was applied to compare the read counts information at each genomic region across all the available isolates. Finally, a Bayesian model was used to convert the differences in read counts to an integer representing copy number variations.

Genome annotation

Annotation of the assembled *L. tropica* genomes was performed using the "COMPANION" online server designed for the annotation and analysis of eukaryotic genomes using *L. major* Friedlin as a reference [46]. COMPANION first reorders and orients the input sequences against ABACAS2 (https://github.com/sanger/sanger-pathogens/ABACAS2) and then uses both homology based RATT [47] and ab initio SNAP [48] and AUGUSTUS [49] techniques for annotations. For protein coding genes functional annotations were defined by OrthoMCL [50] and putative functions are assigned by Pfam-A [51]. Pseudogenes are annotated based on protein-DNA alignments that allow frame-shifts [52] using LAST [53]. tRNAs and rRNAs are annotated using ARAGORN [54] and INFENERAL [55], respectively. Other ncRNAs were assigned using Rfam database [56]. COMPANION also eliminates the overproduction of genes that were transcribed as polycistrons [56]. REVIGO was used to summarize and visualize gene ontology terms [57].

Microsatellite typing

Multilocus microsatellite typing (MLMT) was performed in silico by blasting the forward and reverse primer sequences on each genome using the BioNumerics software version 7.6.1 (Applied Maths, Sint-Martens-Latem, Belgium). In total, 12 independent microsatellite genetic markers specific to L. tropica (GA1, GA2, GA6, GA9n, LIST7010, LIST7011, LIST7027, LIST7033, LIST7039, LIST7040, 4GTG and 27GTGn) were used as previously described by Krayter and colleagues [13,14,15] and for eight L. aethiopica isolates [29]. Bayesian, distancebased and factorial correspondence analyses revealed two populations: India/Asia and Israel/ Palestine that were subdivided, respectively, into two and three subpopulations [14]. A third population, Africa/Galilee, was only supported by the Bayesian analysis [13,15]. Up to four mismatches were allowed for all the reverse primers and up to two mismatches were allowed for the forward primers except for LIST7011 and LIST7039 where up to six mismatches were allowed for their corresponding reverse primers. This led to the identification of all 12 microsatellite across the 19 strains with genome sequence data. The obtained microsatellite profiles were compared to those of 164 previously typed strains of L. tropica and eight L. aethiopica from various geographical locations that were isolated from human cases of CL and VL, hyraxes and sandfly vectors [13]. A few strains with genome sequence data were already analysed by Krayter and colleagues (K112, K26, LRC-L747 and L590) [14, 29] and these represented positive controls for our *in silico* approach for the analyses all 19 strains with genome sequence data.

The methodology used by Krayter and colleagues [14] was followed and the fragment lengths were normalized to that of MHOM/PS/2001/ISL590 (Isolate L590). The obtained data

was compared with the results of 164 typed *L. tropica* and eight *L. aethiopica* isolates [14,15,29].

Phylogenetic analysis

POPTREE2 was used to compute distance measures from allele frequency data and construct a phylogenetic tree from the microsatellite data by using the neighbour-joining (NJ) method with 1,000 bootstrap pseudo replicates and D_{sw} distance measure [58]. The generated tree file in Newick format was visualized and decorated using iTOL (Interactive Tree of Life) [59].

Results

Genome annotation

WGS data of two recent (2014) clinical *L. tropica* isolates representing the outbreak of CL in Lebanon and one historical isolate from Azerbaijan (1974) were generated for this study. A total of 19 *L. tropica* genomes were analysed from nine different countries covering the Middle East and North Africa region (MENA), the Indian subcontinent and one isolate originating from Transcaucasia (Fig 1 and S1 Table). All isolates were inferred to possess 36 chromosomes. On average, 8,133 genes were annotated per isolate with 8,020 coding genes, 455 pseudogenes and a 60% overall GC content (S1 Table).

Microsatellite analysis

The obtained microsatellite profiles for the 19 strains with WGS data were compared to those of 164 previously typed strains of *L. tropica* from various geographical locations that were



Fig 1. Map illustrating the geographic distribution of the analysed isolates of *L. tropica* **and related lineages.** The distribution of human CL caused by *L. tropica*, *L. killicki and L. aethiopica*, with the latter two considered to belong to the *L. tropica* complex [60] is shown as previously reported [2,7]. Brown: countries with CL caused by *L. tropica (sensu stricto)*; yellow: countries with CL caused by both *L. killicki* and *L. tropica;* orange: countries with CL caused by both *L. aethiopica* and *L. tropica*. *L. tropica* isolates with genome sequence data analysed in this study are shown in red circles. The surface of the circles relates to the number of isolates derived from a given country. The three new genomes derived from this study are indicated in red letters (LT1, LT2 and ATCC50129).

https://doi.org/10.1371/journal.pntd.0008684.g001



Fig 2. Phylogenetic relationship between *L. tropica* **strains based on the proportion of shared alleles of microsatellite data.** Bootstrap values > 60% are indicated on the branches. Red squares indicate isolates with corresponding WGS data analysed in this study. The three main populations are highlighted by background colours: Israel/Palestine (red); Africa/Galilee (green) and Asia/India (blue), as previously described [14,15,29]. Individual isolates originating from the Asian continent are depicted with a light green circle, those originating from the African continent are depicted with an orange circle. Green squares refer to *L. aethiopica* isolates, all from Africa [29].

https://doi.org/10.1371/journal.pntd.0008684.g002

isolated from human cases of CL and VL, hyraxes or sandfly vectors [14–15] as well as eight *L. aethiopica* isolates being part of the *L. tropica* species complex [29] (Fig 2). The 19 isolates analysed in this study belonged to three MLMT defined populations that correlated broadly with their geographical distribution. The isolates E50, LRC-L747 and L590 (Israel) clustered among the Israel/Palestine population, as expected [15]. Isolates Kubba (Syria); Rupert, Azad and

KK27 (Afghanistan); Ackerman and LRC-L810 (Israel); Melloy and Boone (Saudi Arabia), KK27 and Azad, (Afghanistan), MN-11, Ma-37 (Jordan); LT1 and LT2 (Lebanon) and Ltr16 (Morocco) were all part of the Africa/Galilee population and were closely related to isolates from various geographic origins including Turkey, Israel and Morocco. In contrast, isolates K112 and K26 (both India) and ATCC50129 (Azerbaijan) clustered among the Asia/India population (Fig 2, S1 Fig, S2 Fig, and S3 Table).

Single nucleotide polymorphism analyses

According to the PCA plot on the obtained SNPs, isolates were divided into five distinct populations/groupings that were considered for subsequent analyses: Population I (MN-11, MA-37, Ltr16); Population II (E50, LRC-L747, LRC-L810); Population III (Melloy, Ackerman, Boone, K26_1), Population IV (K112_1, Rupert, KK27, Kubba, LT1, LT2, Azad) and Population V (ATCC50129) (Fig 3 and S3 Fig). These SNPs based populations do not fully correlate with the geographical distribution of the isolates and are also in contrast with the microsatellite-based analyses in which, for instance, K26_1, K112_1 and ATCC50129 cluster together, consistent with an Asia/India population (Fig 2 and S1 Fig).



Fig 3. Principal component analysis (PCA) based on SNPs of 18 *L. tropica* isolates. The isolates are coloured by country as indicated in the legend and grouped by populations as defined by their PCA based clustering (Populations I to V). MLMT-based populations are added next to isolates' names as follows: A/G for Africa/Galilee, A/I for Asia/India and I/P for Israel/Palestine (See Fig 2 and S3 Fig).

https://doi.org/10.1371/journal.pntd.0008684.g003

Table 2. Summary of the numbers of synonymous, misse	ense and nonsense SNPs, and their averages, in the isolates across five populations defined by their SNPs
(I-V). *: Isolates sequenced in this study. MLMT-based pop	pulations are added next to isolates' names as follows: A/G for Africa/Galilee, A/I for Asia/India and I/P for
Israel/Palestine.	

Isolate	Synonymous	Missense	Nonsense	PCA Population	MLMT
Ltr16	30,016	32,412	61	Ι	
Ma-37	29,937	30,743	63		A/G
MN-11	31,034	32,120	71		
Average	30,329	31,758	65		
E50	54,770	55,038	123	II	I/P
LRC-L747	54,908	55,161	122		
LRC-L810	49,129	52,406	106		A/G
Average	52,936	54,202	117		
Boone	50,632	52,654	105	III	A/G
Melloy	49,434	51,431	101		
Ackerman	50,865	52,886	104		
K26_1	51,599	53,692	106		A/I
Average	50,633	52,666	104		
K112_1	55,995	57,809	110	IV	A/I
Azad	55,702	57,494	111		A/G
KK27	56,006	57,874	118		
LT1*	55,646	57,599	115		
LT2*	53,832	55,143	101		
Rupert	55,927	57,730	117		
Kubba	55,340	57,236	116		
Average	55,493	57,269	113		
ATCC50129*	28,705	28,542	59	V	A/I

https://doi.org/10.1371/journal.pntd.0008684.t002

Highest SNPs heterozygosity was observed in population III and lowest in population I (Table 1). Population IV showed on average the highest number of synonymous and nonsynonymous mutations, population II showed the highest number on average of nonsense mutations (Table 2 and S2 Table).

The relationship between the isolates based on whole-genome SNP pairwise analysis is shown in Fig 4. Isolates belonging to population II clustered together. Unlike the chromosomal aneuploidy profiles (see below), SNPs population analysis divided the isolates into large groups based on geographical distribution, with the exception of isolates originating from Israel that appeared to be divided into two distinct groups. The obtained clustering is essentially in accordance, as expected, with the three main SNPs defined subpopulations reported by Iantorno and colleagues [18] regarding the previously reported 14 isolates. The only exception is for K26_1, which is part of population 2 in [18] but cluster with representatives of population 1 in the expanded analysis across 18 isolates (Fig 4).

GO enrichment analysis

In order to identify potential unique markers among protein coding genes found in one population but not in the others, unique SNPs were investigated among the PCA defined populations I-V. A total of 69, 233, 27 and 50 protein coding genes with population specific SNPs were identified among populations I, II, III and IV (S4 Table). None were found to be unique to ATCC5019 in population V. Gene Ontology enrichment analysis of the unique SNPs containing genes indicated that the corresponding proteins are involved in various biological processes (Fig 5 and S4 Table). Notably, however, nearly 50% of these genes (188 over 379) were





https://doi.org/10.1371/journal.pntd.0008684.g004

annotated as hypothetical proteins and these are not illustrated in <u>Fig 5</u>, which is based on GO term functional annotations.

Unique genes with annotations in population I were involved in various biological processes including N-acetylglucosamine metabolism and other metabolic pathways, regulation of autophagy and ribosome biogenesis. These genes are characterised by relative lower mean numbers of synonymous (dS) and non-synonymous (dN) SNPs per ORF (mean dS +dN = 2.9) compared to the overall mean across all genes and strains (dN+dS = 11.1). Unique genes in





https://doi.org/10.1371/journal.pntd.0008684.g005

population II were involved in protein ubiquitination and tRNA methylation among other processes. These genes were characterised by relatively higher mean number of SNPs per ORF (mean dS+dN = 6.1) compared to ORF with SNPs specific to population I. Unique genes in population III were mainly involved in the tricarboxylic acid cycle, cellular respiration, isoprenoid biosynthesis and antibiotic metabolic process. These genes are characterised by relatively higher mean number of SNPs per ORF (mean dS+dN = 7.7) compared to ORFs from populations I and II. Unique genes in population IV were mainly involved in lipid and glycoprotein biosynthetic processes among others. These genes are characterised by relatively higher number of SNPs per ORF (mean dS+dN = 6.7) compared to population I.

Similar diversity was noted when the 'molecular functions' of the unique SNPs containing genes were analysed. Unique SNPs in population I were concentrated mainly in genes annotated to mediate endoribonuclease activity. Unique SNPs in population II were concentrated mainly with genes annotated to function as serine O-acyltransferase and cyclosporin A binding protein (cyclophilin 15). The molecular functions of unique SNPs in population III were annotated as hydroxymethylglutaryl-CoA synthase among others. Unique SNPs in population IV were concentrated in genes annotated to be involved in sedoheptulose-7-phosphate:D- glyceraldehyde-3-phosphate glyceronetransferase activity and cyclosporin A binding by cyclophilin 2. Alignment of cyclophilin 2 sequence derived from population IV (Isolate Rupert) with that of L590 revealed a single $G \rightarrow A$ SNP leading to a single R163H amino acid substitution.

All genes with populations specific SNPs were characterised by similar mean dN/dS ratios across population (range 0.9–1.4) (S4 Table), which are similar to the overall mean dN/dS value for all ORFs across all 18 isolates (dN/dS = 1.05; range 1–1.3) (S2 Table). Investigating individual genes listed in S4 Table identified several ORF with higher number of SNPs (dN + dN \geq 9) and that were also characterised by higher dN/dS ratios. In population II, the gene LTRL590_010009000 "hypothetical protein, 2C conserved" was characterised by relatively higher dN/dS values (range 1.8–2). In population III, the gene LTRL590_360057000 "hypothetical protein, 2C conserved" among the three isolates was characterised by a more modest dN/dS values (1.3 in all cases). In population IV, the gene LTRL590_340009700 "hypothetical protein, 2C conserved" was also characterised by relatively high dN/dS values across all seven isolates (range 1.8–2.1).

More generally large variations in patterns of dN+dS and dN/dS values were observed across *L. tropica* genes and isolates (S5 Table) with a selection of examples listed in Table 3. Such variations are consistent with *L. tropica* genes experiencing various selective forces and of various strengths, including purifying selection and positive selection. Evidence for positive selection was published for ~20 genes in other *Leishmania* species including *L. donovani* [21]. Some genes with evidence for positive selection in *L. donovani* [21] were also characterised by patterns consistent with positive selection among *L. tropica* isolates (Table 3 and S5 Table). In contrast some *L. tropica* homologues corresponding to genes with positive selection in *L. donovani* [21] are characterised by low variation of dN and dS counts between isolates from the different populations and low dN/dS values, consistent with experiencing purifying selection (e.g. Table 3).

Aneuploidy

Normalized read depth showed that isolates were almost all disomic as expected [18]. Chromosome 31 was present in a trisomic or in a tetrasomic state in all isolates. Chromosome 2 in isolate ATCC50129 was found in a monosomic state. Isolates LT1 and LT2 had a more similar chromosome somy to isolates collected from India (K26_1) and Saudi Arabia (Melloy), respectively. Chromosome somy was not in accordance with the assigned populations (Fig 3) nor with the geographical distribution of the isolates and MLMT data (Fig 6 and S6 Table).

Since chromosome 31 was polysomic in most isolates, we further investigated the annotation of its protein coding genes. Chromosome 31 encodes 412 protein coding genes in L590. These included 177 conserved hypothetical proteins and 49 hypothetical proteins (S7 Table). Most of the genes with functional annotation were involved in mitochondrial RNA metabolic process, phosphatidylcholine metabolic process, mitochondrial gene expression, icosanoid biosynthetic process, prostanoid biosynthetic process among others.

Gene copy number variations

Gene copy number variations (CNVs) among protein coding genes compared to the reference genome L590 were detected as expected [18]. Low copy genes with CNV < 1.5 were more prevalent in population IV with seven isolates in this population exhibiting CNV <1.5. In all isolates an increase in CNVs > 2 for some genes (1–35 genes) was observed. The five populations showed a significant overall difference in their CNVs with the highest number of polyploidy genes on average detected in population I (average = 19) and lowest in population IV

Table 3. Selection of genes illustrating various patterns of SNPs (dN and dS) distribution across the 18 isolates indicating diverse evolutionary forces acting on ORF across *L. tropica* populations. *The comparison of the mean dN+dS between population II (the reference genome of isolate L590 is more closely related to these isolates, <u>S1 Fig</u>) and the population I, III and IV (all with three or more insolates) highlight genes with distinct trends between these populations. This suggests differential evolutionary forces and strength act on these genes across these populations. The genes are ranked according to the dN/dS ratio for genes from LT2. The bottom four genes illustrate ORF experiencing purifying selection among the analysed *L. tropica* isolates and include two genes that have, in contrast, evidence for positive selection in *L. donovani* [21].

Transcript id	Product Description	Ltr16 (POP-I) dN/dS	LRC-L747 (POP-II) dN/dS	LT2 POP-IV dN/dS	Difference mean dN+dS POPII/I*	Difference mean dN+dS POPII/III*	Difference mean dN+dS POPII/IV*	Difference mean dN+dS POPII/V*	Source of the selection
LTRL590_310009200.1	Amastin, putative	18.0	5.3	9.5	1.0	-11.2	-2.2	6.3	This study
LTRL590_340016700.1	Amastin-like surface protein-like protein	2.0	5.0	5.5	10.3	6.3	-0.4	10.3	This study
LTRL590_300017000.1	IQ calmodulin-binding motif containing protein, putative	6.6	2.3	3.3	45.7	8.7	18.4	53.7	[21] <u>S7</u> <u>Table</u>
LTRL590_300021900.1	Hypothetical protein, conserved	2.5	1.0	3.0	-2.3	-4.2	-2.5	1.3	[21] S7 <u>Table</u>
LTRL590_200017700.1	Cysteine peptidase, Clan CA, family C2, putative	4.0	3.0	2.7	0.3	-1.0	0.6	0.0	[21] S6 Table
LTRL590_270033000.1	Hypothetical protein, conserved	1.7	4.2	2.5	5.3	8.1	5.8	16.3	[21] S6 Table
LTRL590_140010500.1	Amastin surface glycoprotein, putative	5.0	4.0	2.3	-0.7	-4.7	-2.9	-1.7	This study
LTRL590_120006700.1	Hypothetical protein, conserved	5.0	7.0	2.3	1.0	-4.7	-2.4	3.3	[21] S6 Table
LTRL590_340012700.1	Flagellar attachment zone protein, putative	1.4	1.4	2.0	6.3	5.8	2.5	12.0	[21] S7 Table
LTRL590_160022000.1	Hypothetical protein	4.0	0.5	2.0	-3.0	-3.2	-0.4	-1.7	[21] S7 Table
LTRL590_140018700.1	Hypothetical protein	1.0	0.4	2.0	2.7	3.0	2.6	2.0	[21] <u>S7</u> Table
LTRL590_340009700.1	Hypothetical protein, conserved (SNPs PCA— Pop IV—Fig 3)	3.0	1.8	1.8	18.0	-2.4	3.2	28.3	This study
LTRL590_150007800.1	Hypothetical protein, conserved	2.7	1.5	1.4	19.3	-10.0	-0.6	28.0	This study
LTRL590_140018600.1	Kinesin K39, putative	1.7	2.0	1.4	1.7	8.7	7.9	20.7	[21] S7 Table
LTRL590_340030000.1	Hypothetical protein, conserved	1.3	1.7	1.4	7.7	5.2	0.9	9.7	[21] <u>S7</u> Table
LTRL590_020006000.1	Phosphatidylinositol kinase related protein, putative	1.2	1.0	1.4	-26.0	-36.3	-15.4	44.7	[<u>21] S7</u> <u>Table</u>
LTRL590_040009100.1	Cysteine peptidase, Clan CA, family C2, putative	1.7	0.8	1.3	4.0	-3.7	0.0	7.3	[21] S6 Table
LTRL590_360057000.1	Hypothetical protein, conserved (SNPs PCA— Pop III—Fig 3)	0.7	1.2	1.3	6.7	2.7	3.5	3.7	This study
LTRL590_190007400.1	Hypothetical protein, conserved	1.3	1.6	1.1	-15.0	-13.3	-6.6	22.7	This study
LTRL590_000009800.1	Hypothetical protein, conserved	0.7	1.0	1.0	-0.3	0.4	-1.8	-3.3	[21] <u>S6</u> Table
LTRL590_140018500.1	Kinesin K39, putative	0.9	0.0	0.8	-7.0	-5.8	-5.0	3.7	[21] S7 Table
LTRL590_000023100.1	Amastin surface glycoprotein, putative	2.0	0.5	0.8	3.0	3.0	0.0	3.0	This study

(Continued)

Transcript id	Product Description	Ltr16 (POP-I) dN/dS	LRC-L747 (POP-II) dN/dS	LT2 POP-IV dN/dS	Difference mean dN+dS POPII/I*	Difference mean dN+dS POPII/III*	Difference mean dN+dS POPII/IV*	Difference mean dN+dS POPII/V*	Source of the selection
LTRL590_220017300.1	Hypothetical protein, conserved	0.8	2.0	0.7	1.3	-1.3	0.0	-0.3	[21] S7 Table
LTRL590_340015200.1	Amastin-like protein	0.6	0.7	0.5	1.3	1.1	1.4	2.3	This study
LTRL590_210021200.1	Hypothetical protein, conserved	0.5	0.3	0.5	1.7	1.7	1.7	1.7	This study
LTRL590_140018200.1	Kinesin, putative	0.8	1.0	0.4	-9.3	3.1	-3.2	5.3	[21] <u>S7</u> Table
LTRL590_360030100.1	Vacuolar protein sorting-associated protein 45- like protein	0.0	0.2	0.3	8.7	3.4	3.6	3.7	[<u>21] S7</u> <u>Table</u>
LTRL590_300009600.1	Alpha/beta hydrolase family, putative	0.0	0.2	0.3	2.0	-1.3	0.2	-1.3	[21] <u>S7</u> Table
LTRL590_240028800.1	Ubiquitin-conjugating enzyme E2, putative	0.0	0.0	0.0	0.0	0.0	0.0	0.0	This study
LTRL590_200006300.1	NADH-ubiquinone oxidoreductase complex I subunit, putative	0.0	0.3	0.0	2.3	2.3	2.3	2.3	This study

Table 3. (Continued)

https://doi.org/10.1371/journal.pntd.0008684.t003

(average = 4) (Table 4). Unique and common genes showing CNV>2 in each isolate and in each population (defined in Fig 3) were examined and 9, 32, 7 and 5 genes were unique to populations I, II, III and IV, respectively. None were unique to ATCC50129 (S8 Table). In population I, protein coding genes with CNV > 2 were found on chromosomes 10, 23, 27, 28, 30 and 35 and were mainly involved in sterol and steroid metabolic process. Population II showed the highest number of unique protein coding genes having CNV>2 (n = 32 genes; <u>S8 Table</u>). These were distributed across chromosomes 1, 2, 6, 10, 17, 20, 24, 27, 28 and 33 with the latter having 10 of these genes. These genes are annotated to mediate various metabolic processes. In population III, unique genes with CNV>2 were located on chromosomes 6, 16, 23, 28 and 33 and were annotated to be part of various biological processes mostly acting in energy production. Population IV had the smallest number of unique genes with CNV>2 compared to all other populations. These were located on chromosomes 6, 27, 28 and 33. Their molecular functions included cyclosporin A binding protein (cyclophilin) among others. Of note, cyclophilin 2 had a CNV>2 in isolate LT1. ATCC50129 had a total of 82 protein coding genes with CNV variation distributed over 22 different chromosomes annotated to mediate various biological processes mainly involved in molybdate ion transport.

LT1 and LT2 had 430 and 54 CNVs, respectively. In LT1 CNVs were mainly located on chromosomes 6 (n = 133) and 28 (n = 182). Genes on the latter two chromosomes were analysed separately. Genes on chromosome 6 were annotated to be mainly involved in deoxyribonucleotide metabolic process whereas genes on chromosome 28 were mainly involved in modulation of host nitric oxide-mediated signal transduction. In contrast, CNVs in LT2 were mainly located on chromosomes 28 (n = 9) and 36 (n = 5) and the remaining 40 genes scattered across nine chromosomes. These genes were annotated to be involved in peptidyl-histidine modification and other biological processes.

Transporter genes

A list of 82 genes annotated to encode transporters in L590 was examined as some of these are thought to play important roles in the biology of the parasite including drug resistance [18].



Fig 6. Comparison of aneuploidy profiles of *L. tropica* **isolates.** Chromosome numbers are listed down the right side of the heat map. Across the bottom, the *L. tropica* isolates are indicated. The dendogram on top shows clusters of the isolates based on the similarity of aneuploidy profiles and was generated by comparison of the Euclidean distances between aneuploidy profiles among the isolates (R packages; stats, gplots). Coloured bar indicates populations as defined in Fig 3: population I (dark blue), population II (dark red), population III (green), population IV (purple) and population V (orange). The geographical origins of the isolates are listed on the upper side at the end of the phylogeny branches. MLMT-based populations are also depicted on the upper side at the end of the phylogeny branches as follows: A/G for Africa/Galilee (green), A/I for Asia/India (blue) and I/P for Israel/Palestine (red).

https://doi.org/10.1371/journal.pntd.0008684.g006

Accordingly, isolate LT1 showed a heterozygous deletion (CNV = 1) of two folate/biopterin transporters (LTRL590_060008200 and LTRL590_060018500) and a phosphate transporter (LTRL590_280036800). ATCC50129 showed a homozygous deletion (CNV = 0) of a glucose transporter. As previously reported [18], highest variations among transporter genes were

Isolate	No. of homozygous deletions $(CNV = 0)^1$	No. of heterozygous deletions $(CNV = 1)^2$	$\overline{W} = 1$ ² No. of polyploid genes (CNV>2) ³		MLMT
Ltr16	14	40	35	Ι	
MA-37	12	14	8		A/G
MN-11	8	19	14		
E50	2	19	8	II	I/P
LRC-L747	2	18	9		
LRC-L810	9	20	9		A/G
Boone	1	8	7	III	A/G
Melloy	1	6	3		
Ackerman	2	7	4		
K26_1	2	8	14		A/I
K112_1	0	1	4	IV	A/I
Azad	0	0	5		A/G
KK27	0	1	4		
Rupert	0	0	3		
Kubba	0	1	1		
LT1*	0	31	1		
LT2*	1	41	9		
ATCC50129*	30	34	0	V	A/I

Table 4. Summary of copy number variations (CNVs) in the *L. tropica* isolates. Populations identified according to shared SNPs (I–V) are shown as are MLMT-based populations (A/G for Africa/Galilee, A/I for Asia/India and I/P for Israel/Palestine). *: Isolates sequenced in this study.

¹Complete gene deletions (CNV = 0);

²Deletion of one copy (CNV = 1, rather than 2);

³Greater than two copies of a gene (CNV>2). No.: number.

https://doi.org/10.1371/journal.pntd.0008684.t004

found in isolates K26_1 and MN-11. Non-synonymous mutations were predominant in folate/biopterin transporters in LT1 while none were found in LT2. Most mutations affecting folate/biopterin transporters were synonymous in LT2 (S9 Table).

Discussion

In this study, microsatellite-based typing, SNPs, chromosome ploidy, and CNVs were investigated to gain further insights into the population genetic diversity of *L. tropica* isolates and the relatedness of different genotyping methods. The microsatellites-based phylogeny highlighted a biased sample of lineages with genome sequence data among the Africa/Galilee populations (12 genomes) defined by Krayter and colleagues [13,14,15], whereas the Israel/Palestine and Asian/Indian populations included three entries each only. Notably, only one isolate was obtained from the African continent (Ltr16) highlighting an important sampling gap, considering the potential African origin of *L. tropica* [12,31].

In silico MLMT analysis assigned the 18 isolates with sequenced genomes to three previously defined distinct populations [13–15] as India/Asia, Israel/Palestine and Israel/Palestine, in accordance to their geographical distribution. As the microsatellite markers are distributed over various chromosomes (including chromosomes 2, 3, 14, 16, 26, 31, 32 and 35), sexual reproduction/hybridization might account for some of the observed discrepancies between SNP defined populations and MLMT genotyping. Notably, chromosomes 14, 16, 26 and 35 were previously shown to have high incidences of localized intraspecific genetic diversity and evidence of hybridization [18]. This has been demonstrated to result in dramatic structural genomic variation [18] and might explain some of the observed discrepancies identified in Fig

4. Further *L. tropica* WGS sampling and comparative analyses will be required to resolve the origin of these discrepancies.

The PCA analysis based on SNPs separated the 18 *L. tropica* isolates into five distinct populations where isolates from the same geographical region belonged to the same population except for isolates from India and from Israel that belonged in two distinct populations each. Isolates belonging to the same population were isolated from countries not sharing borders including Morocco and Jordan (Population I) and Saudi Arabia, India and Israel (Population II). These isolates could have been acquired while people were visiting other countries where CL is endemic and then imported foreign strains of *L. tropica* after they return as previously suggested [15]. Isolates LT1, LT2 and Kubba collected from Lebanon and Syria, respectively, clustered together under population IV consistent with either a transfer event associated with human displacement of refugees [61–65] crossing the Syrian-Lebanon border or movements of the sandfly vector.

Isolate ATCC50129 collected from Baku (Azerbaijan, 1974) was more distinct from all other isolates, clustering separately and having unique profiles for both SNPs and chromosome somy. Consistent with this distinctiveness, isolate ATCC50129 was also part of a separate line-age within the Asia/India population in the MLMT based phylogeny, with currently no phylogenetically close isolates with WGS data. Furthermore, the potential effect of *in vitro* culturing through time, in addition to geographical separation, might have caused additional genetic mutations to occur in this strain [66]. This highlight a lineage of interest for future WGS sampling.

Population II consisted of three isolates (E50, LRC-L747 and LRC-L810) all collected from Israel, which clustered separately from another isolate from Israel (Ackerman, Population III). Previous multi-locus sequence analysis data also showed that a group of strains isolated in Israel formed a distinct subpopulation within the essentially African population, which are part of the so-called Africa/Galilee population defined by Krayter and colleagues [14,15,29]. This could be explained by the presence of different vectors *P. arabicus* and *P. sergenti* in various foci in that region [2,4]. Moreover, imported strains through travelers and immigrants could also contribute to the observed diversity [15,67]. Additionally, different mammalian hosts (human, rock hyrax-Procavia capensis-and possibly other mammals) associated with different infection cycles could have driven a complex history of diversification of L. tropica, with this species considered to have a potential African origin [12,31]. Our combined analyses further highlight the likely complex history of L. tropica spread to its current known broad geographical range. These issues demonstrate the importance of additional strategic sampling of WGS data to identify genetic traits associated with potential distinct insect-mammalian infection cycles, under consistent *in vitro* conditions and a passage-controlled manner [19]. The distribution of the rock hyrax across Africa and the Middle East [68] could explain the broad distribution of the African/Galilee population of L. tropica, with the rock hyrax considered by some as the main reservoir for *L. tropica* across Africa and the Middle East [68]. The distinct MLMT defined Asian/India lineage, that has isolates derived from regions beyond the distribution of the rock hyrax, might represent genetic variants associated with a distinct set of mammalian hosts.

Similarly, and as expected [18,19], chromosome somy did neither fully correlate to the geographical distribution of the isolates nor with the SNP or MLST assigned populations. This has been previously reported for other *Leishmania* species [44]. Isolates LT1 and LT2 had a more similar chromosome somy to isolates collected from India (K26_1) and Saudi Arabia (Melloy), respectively. The majority of the chromosomes existed in a disomic state. The chromosome numbers in *Leishmania* species vary in response to their environment resulting in mosaic chromosome aneuploidy [69]. The biological relevance, if any, of the unusual somy for several chromosome for ATCC50129 will have to be investigate through further genome sampling across related strains.

Chromosome 31 was present in a trisomic state in almost all isolates as has been previously observed in all Leishmania species so far, including L. tropica [18, 21, 23]. It has been suggested that this chromosome ploidy is needed to expedite iron uptake as genes acting in iron metabolism are enriched on chromosome 31 [69]. Our analysis showed that annotation of protein coding genes encoded on chromosome 31 function in various metabolic processes and mainly included mitochondrial RNA metabolic process, phosphatidylcholine metabolic process, eicosanoid and prostanoid biosynthetic processes. Phosphatidylcholine, a cell membrane phospholipid, which is targeted by miltefosine, a broad spectrum anti-leishmanial drug [70]. Phosphatidylcholine in Leishmania membranes presents long polyunsaturated fatty acids different from those found in mammals. This particular characteristic may aid in the resistance of the parasite to the highly oxidative environment characteristic of the phagolysosomes in macrophages [71]. On the other hand, genes acting in eicosanoids biosynthesis were also annotated on chromosome 31. Eicosanoids are signalling molecules that are key players in the regulation of host immune responses during Leishmania infection [72] and the expression of mediators from the eicosanoid biosynthetic pathway was enriched in CL [73]. The potential effect of chromosome 31 CNVs on specific biosynthetic pathways and modulation of the host immune response merits further experimental investigations.

Gene CNVs were investigated to identify potential unique genes between populations having CNV>2. Genes with CNVs were distributed across various chromosomes. Chromosomes 6, 28 and 33 appeared to be particularly rich with such CNVs. Population I (as defined by SNPs) had also a unique set of genes with CNV>2 that are annotated to mediate cholestenol delta-isomerase activity or an intramolecular oxidoreductase activity, transposing C = Cbonds. To our knowledge no report exists on the cholestenol delta-isomerase activity in Leishmania beyond in silico annotations. Cholestenol delta-isomerase is an enzyme of the sterol biosynthetic pathway. Sterols in eukaryotes serve multiple physiological roles acting in forming structural components of membranes and as precursors to signalling molecules needed for the regulation of growth and development [74]. Population II (as defined by unique SNPs) showed the highest number of unique genes with CNV>2 annotated as mediating coproporphyrinogen oxidase activity or an oxidoreductase activity, acting on the CH-CH group of donors, with oxygen as acceptor. Like most trypanosomatids, Leishmania species are auxotrophic for haem (or heme) and must acquire this essential porphyrin from their hosts [75]. Coproporphyrinogen oxidase is part of the haem biosynthesis pathway and appears to vary between species [76]. Leishmania seem to have acquired the genes encoding the enzymes for the last three steps of the pathway from γ -proteobacteria lateral gene transfers [77]. Although *Leishmania* is able to break down haem to be used as a source of iron [78], the exact function of these enzymes remains unknown [79]. In population III (as defined by unique SNPs), unique genes with CNVs > 2 are annotated to encode functions in metal and iron-sulfur (Fe-S) cluster binding, among other functions. Fe-S clusters are versatile cofactors of numerous proteins and are part of various essential biological processes such as the TCA cycle, redox homeostasis, DNA replication and repair and protein translation [79]. One of the unique genes identified in population IV was cyclophilin 2 that have been shown in other Leishmania species to have a cyclosporin A binding activity [80]. It is well established that cyclosporin A exhibits an inhibitory effect on T cell-mediated immunity [81,82]. Nevertheless, cyclosporin A can also displayed anti-microbial activity against a variety of protozoan pathogens, including Leishmania [83] and was shown to have a suppressive effect on the development of L. tropica-induced lesions in susceptible BALB/c mice [83]. Cyclosporin is produced by fungi such as Tolypocladium inflatum [84]. As fungi have been identified in the gut of sandflies [85], differential

binding of cyclophilin to cyclosporin A might be favoring leishmanial survival in that habitat. The exact role of genes encoding cyclophilins in *L. tropica* biology requires further investigations. The potential functional relevance of the SNPs ($G \rightarrow A$) and CNV>2 associated with cyclophilin 2 of *L. tropica* identified in this study will be interesting to investigate. Similarly, the potential effect of gene CNVs on specific biosynthetic pathways in either the mammalian host or the insect vector will also merit experimental explorations.

This comparative study has further revealed the heterogeneity of L. tropica isolated from various foci scattering into five distinct populations across the broad range of the parasite geographic distribution. Microsatellite analysis clustered the isolates into three populations that correlated broadly with their geographical distribution where LT1 and LT2 (Lebanon) were part of the Africa/Galilee population while ATCC50129 (Azerbaijan) was part of the Asia/ India population. Unique genes that can be attributed to essential biological processes, including enzymes involved in energy metabolism or proteins interactions with drugs, have been found to differentiate *L. tropica* populations. However, a number of genes encoding hypothetical proteins differentiating populations were also identified, further highlighting our knowledge gap in the biology of L. tropica. Notably, some of the gene encoding hypothetical proteins with population specific SNPs where characterised by differential patterns of dN+dS SNPs and dN/dS ratios across the different identified populations (Table 3). These features highlight genes experiencing differential evolutionary dynamic, which could be due for example, to positive selections acting upon them in response to environmental changes, as suggested for genes from other Leishmania species [21]. This could include parasites exposed to different combinations of sandfly vectors and/or mammalian hosts, or more general changes in the environmental conditions due to habitat variations experienced by the insect vectors and mammalian hosts. Consistent with the contrasting biology between L. tropica and L. donovani, some genes with evidence for positive selection in the latter [21] were characterised by patterns of SNPs consistent with purifying selection acting upon L. tropica genes e.g. LTRL590_300009600 encoding a putative alpha/beta hydrolase family member (Table 3). The annotated L. tropica amastin gene family members, encoding immunogenic surface proteins, were also characterised with contrasting patterns of SNPs between genes and the 18 L. tropica isolates (a selection is listed in Table 3), as observed among other *Leishmania* species [21]. This include some L. tropica amastin encoding genes which could be under positive selection in some populations (Table 3). Some transporters encoding genes were also characterised by contrasting patterns of SNPs between the different populations. These included the glucose transporter 2 (LTRL590_360075500) and the folate/biopterin transporter, putative (LTRL590_000024200.1). Various transporters, including ABC transporters, glucose transporters and the folate/biopterin transporter are differentially expressed during the parasite life cycle and have been previously shown to be implicated in drug resistance [21]. The differential patterns of SNPs between population observed in some genes could also have potentially relevant functional implications, which would be of interest to investigate experimentally. Such genes represent attractive targets to gain new insights into the molecular basis of host-parasite interactions in either insect or mammalian, or both, hosts. These could also be exploited to design more accurate typing schemes for more detailed epidemiological analyses to, for instance, differentiate potential zoonotic from anthroponotic infection cycles [68]. Our analyses also further highlight the importance of future strategic sampling of genome sequences across the broad genetic diversity and broad host and geographic range for L. tropica to further expand our understanding of the complex relationship between the parasite tremendous genomic diversity with the parasites' biology, including diverse symptomology among humans.

Supporting information

S1 Fig. Neighbor-joining tree showing the phylogenetic relationship between the 19 *L. tropica* strains analysed in this study based on the proportion of shared alleles of microsatellite data. The NJ tree (1000 bootstrap) was constructed using POPOTREE2 [58] and visualized using iTOL [59]. Bootstrap values are indicated below the branches. The three main populations (as described in [14,15,29]) are indicated in (i) red, Israel/Palestine (I/P), (ii) green, Africa/Galilee (A/G) and (iii) blue, Asia/India (A/I). The three genome sequence data generated in this study are indicated in red. (PDF)

S2 Fig. Neighbor-joining tree rooted with the *L. aethiopica* containing lineage (consistent with a potential African origin for *L. tropica*) constructed based on the proportion of shared alleles of microsatellite data. The NJ tree (1000 bootstrap) was constructed using POPOTREE2 [58] and visualized using iTOL [59]. Red squares indicate isolates with WGS data analysed in this study; green squares, *L. aethiopica* isolates. The three main populations (as described in [14,15,29]) are indicated in (i) red, Israel/Palestine (I/P), (ii) green, Africa/Galilee (A/G) and (iii) blue, Asia/India (A/I). Orange stripes on the right indicate isolates originating from the African continent and green from the Asian continent. (PDF)

S3 Fig. Principal component analysis (PCA) performed on isolates belonging to popula-tions III, IV and V. In order to obtain a better resolution of separate populations and a better separation between various isolates, PCA analysis was repeated on populations III, IV and V. The isolates are coloured by country as indicated in the legend and broadly grouped by populations as defined by their initial PCA based clustering (Populations I to V illustrated in Fig.3). (PDF)

S1 Table. Comparison of the genome contents, geographical origin and genome characteristics. Isolates sequenced in this study are shown in red. ^a coverage = (read count *read length*2) / total genome size; Genome size estimate = 33Mbp; ^b in genes/ megabase; ^c collected from *P. arabicus*; NA: not available. (XLSX)

S2 Table. Genomic variant annotations and functional effect prediction of SNPs across 18 isolates and their annotated genes. SNPs annotations were obtained using the SNPEff software [52]. dS: synonymous and dN: non-synonymous SNPs; 5'UTR and 3'UTR and gained stop codons are also listed.

(XLSX)

S3 Table. MLMT data of the sequenced genomes compared to previously reported data. The chromosome locations of the microsatellites are based on *L. major* as reported by Schwenkenbecher and colleagues [13] and Jamjoom and colleagues [86]. S2 Table was modified from the original "Additional file 1: S1 Table" from Krayter and colleagues [29] by incorporating the *in silico* MLMT typing for the 19 isolates with WGS data (yellow highlights). Entries analysed by both Krayter and colleagues [29] and the WGS data are highlighted in bold red text. Table B lists exclusively the *in silico* data for the 19 isolates with WGS data analysed in this study. (XLSX)

S4 Table. Unique SNPs containing genes specific for either populations I, II, III or IV. (XLSX)

S5 Table. Summary table facilitating comparison of patterns for SNPs on ORF (dN+dS, dN/dS) across all protein coding genes and 18 isolates of *L. tropica*. A subset is illustrated in Table 3.

(XLSX)

S6 Table. Chromosome-somy. MLMT-based populations are added above isolates' names as follows: A/G for Africa/Galilee, A/I for Asia/India and I/P for Israel/Palestine. (XLSX)

S7 Table. Annotation of protein coding genes located on chromosome 31 in L590. (XLSX)

S8 Table. Unique gene having CNV>2 in each SNPs defined population. (XLSX)

S9 Table. List of SNPs present in annotated transporter genes for each isolate. (XLSX)

Author Contributions

Conceptualization: Tamara Salloum, Robert P. Hirt, Sima Tokajian.

Funding acquisition: Robert P. Hirt, Sima Tokajian.

Investigation: Tamara Salloum.

Methodology: Tamara Salloum, Rim Moussa, Ryan Rahy, Jospin Al Deek, Ibrahim Khalifeh, Rana El Hajj, Neil Hall.

Project administration: Robert P. Hirt, Sima Tokajian.

Supervision: Robert P. Hirt, Sima Tokajian.

Validation: Robert P. Hirt, Sima Tokajian.

Visualization: Tamara Salloum, Rim Moussa, Ryan Rahy.

Writing - original draft: Tamara Salloum.

Writing - review & editing: Ibrahim Khalifeh, Neil Hall, Robert P. Hirt, Sima Tokajian.

References

- World Health Organization. WHO Fact sheets, Leishmaniasis, 14 March 2019. <u>https://www.who.int/news-room/fact-sheets/detail/leishmaniasis</u>. Accessed on November 11, 2019.
- 2. Rostamian M, Niknam HM. *Leishmania tropica*: What we know from its experimental models. Adv Parasit. 2019; 104:1–38.
- Zhang CY, Lu XJ, Du XQ, Jian J, Shu L, Ma Y. Phylogenetic and evolutionary analysis of Chinese Leishmania isolates based on multilocus sequence typing. Plos One. 2013; 8:e63124. <u>https://doi.org/ 10.1371/journal.pone.0063124</u> PMID: 23646184
- Schnur LF, Nasereddin A, Eisenberger CL, Jaffe CL, El Fari M, Azmi K, et al. Multifarious characterization of *Leishmania tropica* from a Judean desert focus, exposing intraspecific diversity and incriminating *Phlebotomus sergenti* as its vector. Am J Trop Med Hyg.2004; 70:364–72. PMID: 15100448
- Mebrahtu YB, Lawyer PG, Ngumbi PM, Kirigi G, Mbugua J, Gachihi G, et al. A new rural focus of cutaneous leishmaniasis caused by *Leishmania tropica* in Kenya. Trans R Soc Trop Med Hyg. 1992; 86:381–7. https://doi.org/10.1016/0035-9203(92)90230-a PMID: 1332221
- Schönian G, Schnur L, El Fari M, Oskam L, Kolesnikov AA, Sokolowska-Köhler W, et al. Genetic heterogeneity in the species *Leishmania tropica* revealed by different PCR-based methods. Trans R Soc Trop Med Hyg. 2001; 95:217–24. https://doi.org/10.1016/s0035-9203(01)90173-7 PMID: 11355565

- Cunze S, Kochmann J, Koch LK, Hasselmann KJ, Klimpel S. Leishmaniasis in Eurasia and Africa: geographical distribution of vector species and pathogens. R Soc Open Sci. 2019; 6:190334. <u>https://doi.org/10.1098/rsos.190334</u> PMID: 31218068
- Ghatee MA, Mirhendi H, Karamian M, Taylor WR, Sharifi I, Hosseinzadeh M, et al. Population structures of *Leishmania infantum* and *Leishmania tropica* the causative agents of kala-azar in Southwest Iran. Parasitol Res. 2018; 117:3447–58. https://doi.org/10.1007/s00436-018-6041-1 PMID: 30105405
- Zijlstra EE, Musa AM, Khalil EA, El Hassan IM, El-Hassan AM. Post-kala-azar dermal leishmaniasis. Lancet Infect Dis. 2003; 3:87–98. https://doi.org/10.1016/s1473-3099(03)00517-6 PMID: 12560194
- Ahluwalia S, Lawn SD, Kanagalingam J, Grant H, Lockwood DN. Mucocutaneous leishmaniasis: an imported infection among travellers to central and South America. Bmj. 2004; 329:842–4. <u>https://doi.org/10.1136/bmj.329.7470.842</u> PMID: 15472268
- 11. Ready PD. Biology of phlebotomine sandflies as vectors of disease agents. Ann Rev Entomol. 2013;58.
- Akhoundi M, Kuhls K, Cannet A, Votýpka J, Marty P, Delaunay P, Sereno D. A historical overview of the classification, evolution, and dispersion of *Leishmania* parasites and sandflies. PLoS Negl Trop Dis. 2016; 10:e0004349. https://doi.org/10.1371/journal.pntd.0004349 PMID: 26937644
- Schwenkenbecher JM, Wirth T, Schnur LF, Jaffe CL, Schallig H, Al-Jawabreh A, et al. Microsatellite analysis reveals genetic structure of *Leishmania tropica*. Int J Parasitol. 2006; 36:237–46. <u>https://doi.org/10.1016/j.ijpara.2005.09.010</u> PMID: 16307745
- Krayter L, Alam MZ, Rhajaoui M, Schnur LF, Schönian G. Multilocus Microsatellite Typing reveals intrafocal genetic diversity among strains of *Leishmania tropica* in Chichaoua Province, Morocco. Infect Gen Evol. 2014; 28:233–9.
- Krayter L, Bumb RA, Azmi K, Wuttke J, Malik MD, Schnur LF, et al. Multilocus microsatellite typing reveals a genetic relationship but, also, genetic differences between Indian strains of *Leishmania tropica* causing cutaneous leishmaniasis and those causing visceral leishmaniasis. Parasites Vector. 2014; 7(1):123.
- Schönian G, Lukeš J, Stark O, Cotton JA. Molecular evolution and phylogeny of *Leishmania*. In Drug Resistance in *Leishmania* Parasites 2018 (pp. 19–57). Springer, Cham.
- Aslett M, Aurrecoechea C, Berriman M, Brestelli J, Brunk BP, Carrington M, et al. TriTrypDB: a functional genomic resource for the Trypanosomatidae. Nucleic Acids Res. 2009; 38:D457–62. https://doi. org/10.1093/nar/gkp851 PMID: 19843604
- Iantorno SA, Durrant C, Khan A, Sanders MJ, Beverley SM, Warren WC, et al. Gene expression in Leishmania is regulated predominantly by gene dosage. MBio. 2017; 8:e01393–17. https://doi.org/10. 1128/mBio.01393-17 PMID: 28900023
- Bussotti G, Gouzelou E, Boité MC, Kherachi I, Harrat Z, Eddaikra N, et al. *Leishmania* genome dynamics during environmental adaptation reveal strain-specific differences in gene copy number variation, karyotype instability, and telomeric amplification. MBio. 2018; 9:e01399–18. <u>https://doi.org/10.1128/mBio.01399-18 PMID: 30401775</u>
- Peacock CS, Seeger K, Harris D, Murphy L, Ruiz JC, Quail MA, et al. Comparative genomic analysis of three *Leishmania* species that cause diverse human disease. Nature Gen. 2007; 39:839.
- Downing T, Imamura H, Decuypere S, Clark TG, Coombs GH, Cotton JA, et al. Whole genome sequencing of multiple *Leishmania donovani* clinical isolates provides insights into population structure and mechanisms of drug resistance. Genome Res. 2011; 21:2143–56. <u>https://doi.org/10.1101/gr.</u> 123430.111 PMID: 22038251
- Ivens AC, Peacock CS, Worthey EA, Murphy L, Aggarwal G, Berriman M, et al. The genome of the kinetoplastid parasite, *Leishmania major*. Science. 2005; 309:436–42. https://doi.org/10.1126/science. 1112680 PMID: 16020728
- Rogers MB, Hilley JD, Dickens NJ, Wilkes J, Bates PA, Depledge DP, et al. Chromosome and gene copy number variation allow major structural change between species and strains of *Leishmania*. Genome Res. 2011; 21:2129–42. https://doi.org/10.1101/gr.122945.111 PMID: 22038252
- Fransenn SU, Durrant C, Stark O, Moser B, Downing T, Imamura H, et al. Global genome diversity of the *Leishmania donovani* complex. eLife. 2020; 9:e51243. https://doi.org/10.7554/eLife.51243 PMID: 32209228
- 25. Kazemi B. Genomic organization of Leishmania species. Iran J Parasitol. 2011; 6:1.
- 26. Sterkers Y, Lachaud L, Crobu L, Bastien P, Pagès M. FISH analysis reveals aneuploidy and continual generation of chromosomal mosaicism in Leishmania major. Cellular Microbiol. 2011; 13:274–83.
- Saroufim M, Charafeddine K, Issa G, Khalifeh H, Habib RH, Berry A, et al. Ongoing epidemic of cutaneous leishmaniasis among Syrian refugees, Lebanon. Emerg Infect Dis. 2014; 20:1712. https://doi.org/10.3201/eid2010.140288 PMID: 25279543

- Salloum T, Khalifeh I, Tokajian S. Detection, molecular typing and phylogenetic analysis of *Leishmania* isolated from cases of leishmaniasis among Syrian refugees in Lebanon. Parasite Epidemiol Control. 2016; 1:159–68. https://doi.org/10.1016/j.parepi.2016.02.002 PMID: 29988171
- Krayter L, Schnur LF, Schönian G. The genetic relationship between *Leishmania aethiopica* and *Leishmania tropica* revealed by comparing microsatellite profiles. PloS One. 2015; 10:e0131227. <u>https://doi.org/10.1371/journal.pone.0131227 PMID: 26196393</u>
- Sergiev V, Kondrashin A, Litvinov S, Morozova L, Turbabina N, Stepanova E, et al. Epidemiology and control of leishmaniasis in the former USSR: a review article. Iran J Parasitol. 2018; 13:342. PMID: 30483324
- Momen H, Cupolillo E. Speculations on the origin and evolution of the genus *Leishmania*. Mem Inst Oswaldo Cruz. 2000; 95:583–8. https://doi.org/10.1590/s0074-0276200000400023 PMID: 10904419
- Svobodova M, Votypka J, Peckova J, Dvorak V, Nasereddin A, Baneth G, et al. Distinct transmission cycles of *Leishmania tropica* in 2 adjacent foci, Northern Israel. Emerg Infect Dis. 2006; 12:1860. https://doi.org/10.3201/eid1212.060497 PMID: 17326936
- Ridley DS, Ridley MJ. The evolution of the lesion in cutaneous leishmaniasis. J Pathol. 1983; 141:83– 96. https://doi.org/10.1002/path.1711410109 PMID: 6620010
- El Hajj R, Youness HB, Lachaud L, Bastien P, Masquefa C, Bonnet PA, et al. EAPB0503: An Imiquimod analog with potent in vitro activity against cutaneous leishmaniasis caused by *Leishmania major* and *Leishmania tropica*. PLoS Neglected Trop Dis. 2018; 12:e0006854.
- Andrews S. FastQC: a quality control tool for high throughput sequence data. Available online at: http://www.bioinformatics.babraham.ac.uk/projects/fastqc
- Bolger AM, Lohse M, Usadel B. Trimmomatic: a flexible trimmer for Illumina sequence data. Bioinformatics. 2014; 30:2114–20. https://doi.org/10.1093/bioinformatics/btu170 PMID: 24695404
- Langmead B, Salzberg SL. Fast gapped-read alignment with Bowtie 2. Nat. Methods. 2012; 9:357. https://doi.org/10.1038/nmeth.1923 PMID: 22388286
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The sequence alignment/map format and SAMtools. Bioinformatics. 2009; 25:2078–9. https://doi.org/10.1093/bioinformatics/btp352 PMID: 19505943
- Garrison E, Marth G. Haplotype-based variant detection from short-read sequencing. arXiv preprint arXiv. 2012;207.3907.
- Li H. Tabix: fast retrieval of sequence features from generic TAB-delimited files. Bioinformatics. 2011; 27:718–9. https://doi.org/10.1093/bioinformatics/btq671 PMID: 21208982
- Danecek P, Auton A, Abecasis G, Albers CA, Banks E, DePristo MA, et al. The variant call format and VCFtools. Bioinformatics. 2011; 27:2156–8. <u>https://doi.org/10.1093/bioinformatics/btr330</u> PMID: 21653522
- 42. Zheng X. A Tutorial for the R Package SNPRelate. University of Washington, Washington, USA. 2013 Jun 8.
- 43. Cingolani P, Platts A, Wang LL, Coon M, Nguyen T, Wang L, et al. A program for annotating and predicting the effects of single nucleotide polymorphisms, SnpEff: SNPs in the genome of Drosophila melanogaster strain w1118; iso-2; iso-3. Fly. 2012; 6:80–92.
- 44. Zackay A, Cotton JA, Sanders M, Hailu A, Nasereddin A, Warburg A, et al. Genome wide comparison of Ethiopian *Leishmania donovani* strains reveals differences potentially related to parasite survival. PLoS Gen. 2018; 14:e1007133.
- **45.** Klambauer G, Schwarzbauer K, Mayr A, Clevert DA, Mitterecker A, Bodenhofer U, et al. cn. MOPS: mixture of Poissons for discovering copy number variations in next-generation sequencing data with a low false discovery rate. Nucleic Acids Res. 2012;40:e69-.
- 46. Steinbiss S, Silva-Franco F, Brunk B, Foth B, Hertz-Fowler C, Berriman M, et al. Companion: a web server for annotation and analysis of parasite genomes. Nucleic Acids Res. 2016; 44:W29–34. <u>https://doi.org/10.1093/nar/gkw292 PMID: 27105845</u>
- Otto TD, Dillon GP, Degrave WS, Berriman M. RATT: rapid annotation transfer tool. Nucleic Acids Res. 2011; 39(9):e57–. https://doi.org/10.1093/nar/gkq1268 PMID: 21306991
- Korf I. Gene finding in novel genomes. BMC Bioinformatics. 2004; 5:59. <u>https://doi.org/10.1186/1471-2105-5-59 PMID: 15144565</u>
- Stanke M, Tzvetkova A, Morgenstern B. AUGUSTUS at EGASP: using EST, protein and genomic alignments for improved gene prediction in the human genome. Genome Biol. 2006; 7:S11. <u>https://doi.org/10.1186/gb-2006-7-s1-s11</u> PMID: 16925833
- Li L, Stoeckert CJ, Roos DS. OrthoMCL: identification of ortholog groups for eukaryotic genomes. Genome Res. 2003; 13:2178–89. https://doi.org/10.1101/gr.1224503 PMID: 12952885

- Finn RD, Bateman A, Clements J, Coggill P, Eberhardt RY, Eddy SR, et al. Pfam: the protein families database. Nucleic Acids Res. 2013; 42:D222–30. https://doi.org/10.1093/nar/gkt1223 PMID: 24288371
- Sheetlin SL, Park Y, Frith MC, Spouge JL. Frameshift alignment: statistics and post-genomic applications. Bioinformatics. 2014; 30:3575–82. https://doi.org/10.1093/bioinformatics/btu576 PMID: 25172925
- Kiełbasa SM, Wan R, Sato K, Horton P, Frith MC. Adaptive seeds tame genomic sequence comparison. Genome Res. 2011; 21:487–93. https://doi.org/10.1101/gr.113985.110 PMID: 21209072
- Laslett D, Canback B. ARAGORN, a program to detect tRNA genes and tmRNA genes in nucleotide sequences. Nucleic Acids Res. 2004; 32:11–6. https://doi.org/10.1093/nar/gkh152 PMID: 14704338
- Nawrocki EP, Eddy SR. Infernal 1.1: 100-fold faster RNA homology searches. Bioinformatics. 2013; 29:2933–5. https://doi.org/10.1093/bioinformatics/btt509 PMID: 24008419
- 56. Nawrocki EP, Burge SW, Bateman A, Daub J, Eberhardt RY, Eddy SR, et al. Rfam 12.0: updates to the RNA families database. Nucleic Acids Res. 2014; 43:D130–7. <u>https://doi.org/10.1093/nar/gku1063</u> PMID: 25392425
- Supek F, Bošnjak M, Škunca N, Šmuc T. REVIGO summarizes and visualizes long lists of gene ontology terms. PloS one. 2011; 6:e21800. https://doi.org/10.1371/journal.pone.0021800 PMID: 21789182
- Takezaki N, Nei M, Tamura K. POPTREE2: Software for constructing population trees from allele frequency data and computing other population statistics with Windows interface. Mol Biol Evol. 2009; 27:747–52. https://doi.org/10.1093/molbev/msp312 PMID: 20022889
- Letunic I, Bork P. Interactive Tree Of Life (iTOL): an online tool for phylogenetic tree display and annotation. Bioinformatics. 2006; 23:127–8. https://doi.org/10.1093/bioinformatics/btl529 PMID: 17050570
- Chaara D, Ravel C, Bañuls AL, Haouas N, Lami P, Talignani L, et al. Evolutionary history of *Leishmania* killicki (synonymous *Leishmania tropica*) and taxonomic implications. Parasites & vectors. 2015; 8:198.
- Saleh S, Farah A, Dimassi H, El Arnaout N, Constantin J, Osman M, et al. Using mobile health to enhance outcomes of noncommunicable diseases care in rural settings and refugee camps: randomized controlled trial. JMIR mHealth uHealth. 2018; 6:e137. https://doi.org/10.2196/mhealth.8146 PMID: 30006326
- Saleh S, Alameddine M, Farah A, El Arnaout N, Dimassi H, Muntaner C, et al. eHealth as a facilitator of equitable access to primary healthcare: the case of caring for non-communicable diseases in rural and refugee settings in Lebanon. Int J Public Health. 2018; 63:577–88. <u>https://doi.org/10.1007/s00038-018-1092-8 PMID: 29546440</u>
- **63.** Premkumar A, Raad K, Haidar MH. Rethinking the social history in the era of biolegitimacy: global health and medical education in the care of Palestinian and Syrian refugees in Beirut, Lebanon. Anthropol Med. 2016; 23:14–29. https://doi.org/10.1080/13648470.2015.1135785 PMID: 26982432
- Ammar W, Kdouh O, Hammoud R, Hamadeh R, Harb H, Ammar Z, et al. Health system resilience: Lebanon and the Syrian refugee crisis. J Glob Health. 2016; 6(2).
- Fakhoury T. Multi-level governance and migration politics in the Arab world: The case of Syria's displacement. J Ethn Migr Stud. 2019; 45:1310–26.
- 66. Kuhls K, Keilonat L, Ochsenreither S, Schaar M, Schweynoch C, Presber W, et al. Multilocus microsatellite typing (MLMT) reveals genetically isolated populations between and within the main endemic regions of visceral leishmaniasis. Microbes Infect. 2007; 9:334–43. <u>https://doi.org/10.1016/j.micinf.</u> 2006.12.009 PMID: 17307010
- Solomon M, Greenberger S, Baum S, Pavlotsky F, Barzilai A, Schwartz E. Unusual forms of cutaneous leishmaniasis due to *Leishmania major*. J Eur Acad Dermatol. 2016; 30:1171–5.
- Waitz Y, Paz S, Meir D, Malkinson D. Effects of land use type, spatial patterns and host presence on Leishmania tropica vectors activity. Parasit Vectors. 2019; 12:320. https://doi.org/10.1186/s13071-019-3562-0 PMID: 31238938
- Valdivia HO, Reis-Cunha JL, Rodrigues-Luiz GF, Baptista RP, Baldeviano GC, Gerbasi RV, et al. Comparative genomic analysis of *Leishmania (Viannia) peruviana* and *Leishmania (Viannia) braziliensis*. BMC Genomics. 2015; 16:715. https://doi.org/10.1186/s12864-015-1928-z PMID: 26384787
- 70. Rakotomanga M, Blanc S, Gaudin K, Chaminade P, Loiseau PM. Miltefosine affects lipid metabolism in Leishmania donovani promastigotes. Antimicrob Agents Chemother. 2007; 51:1425–30. <u>https://doi.org/10.1128/AAC.01123-06 PMID: 17242145</u>
- Zhang K, Beverley SM. Phospholipid and sphingolipid metabolism in *Leishmania*. Mol Biochem Parasitol. 2010; 170:55–64. https://doi.org/10.1016/j.molbiopara.2009.12.004 PMID: 20026359
- 72. Daugschies A, Joachim A. Eicosanoids in parasites and parasitic infections. Adv Parasitol. 2000; 46:181–240. https://doi.org/10.1016/s0065-308x(00)46009-4 PMID: 10761556

- 73. França-Costa J, Andrade BB, Khouri R, Van Weyenbergh J, Malta-Santos H, da Silva Santos C, et al. Differential expression of the eicosanoid pathway in patients with localized or mucosal cutaneous leishmaniasis. J Infect Dis. 2015; 213:1143–7. https://doi.org/10.1093/infdis/jiv548 PMID: 26582954
- 74. Fügi MA, Gunasekera K, Ochsenreiter T, Guan X, Wenk MR, Mäser P. Genome profiling of sterol synthesis shows convergent evolution in parasites and guides chemotherapeutic attack. J Lipid Res. 2014; 55:929–38. https://doi.org/10.1194/jlr.M048017 PMID: 24627128
- **75.** Opperdoes FR, Coombs GH. Metabolism of *Leishmania*: proven and predicted. Trends in Parasitol. 2007; 23:149–58.
- 76. Franken AC, Werner ER, Haas H, Lokman BC, van den Hondel CA, Ram AF, et al. The role of coproporphyrinogen III oxidase and ferrochelatase genes in heme biosynthesis and regulation in *Aspergillus niger*. Appl Microbiol Biotechnol. 2013; 97:9773–85. https://doi.org/10.1007/s00253-013-5274-2 PMID: 24113826
- 77. Kořený L, Lukeš J, Oborník M. Evolution of the haem synthetic pathway in kinetoplastid flagellates: an essential pathway that is not essential after all?. Int J Parasitol. 2010; 40:149–56. <u>https://doi.org/10.1016/j.ijpara.2009.11.007</u> PMID: 19968994
- Carvalho S, Cruz T, Santarém N, Castro H, Costa V, Tomás AM. Heme as a source of iron to *Leishmania infantum* amastigotes. Acta tropica. 2009; 109:131–5. https://doi.org/10.1016/j.actatropica.2008. 10.007 PMID: 19013419
- 79. Kumar A, Chauhan N and Singh S. Understanding the Cross-Talk of Redox Metabolism and Fe-S Cluster Biogenesis in *Leishmania* Through Systems Biology Approach. Front Cell Infect Microbiol. 9:15. https://doi.org/10.3389/fcimb.2019.00015 PMID: 30778378
- Yau WL, Blisnick T, Taly JF, Helmer-Citterich M, Schiene-Fischer C, Leclercq O, et al. Cyclosporin A treatment of *Leishmania donovani* reveals stage-specific functions of cyclophilins in parasite proliferation and viability. PLoS Neglected Trop Dis; 4:e729.
- Walsh CT, Zydowsky LD, McKeon FD. Cyclosporin A, the cyclophilin class of peptidylprolyl isomerases, and blockade of T cell signal transduction. J Biol Chem. 1992; 267:13115–8. PMID: 1618811
- Ho S, Clipstone N, Timmermann L, Northrop J, Graef I, Fiorentino D, et al. The mechanism of action of cyclosporin A and FK506. Clin Immunol Immunopathol. 1996; 80:S40–5. <u>https://doi.org/10.1006/clin.</u> 1996.0140 PMID: 8811062
- Solbach W, Forberg K, Kammerer E, Bogdan C, Röllinghoff M. Suppressive effect of cyclosporin A on the development of *Leishmania tropica*-induced lesions in genetically susceptible BALB/c mice. J Immunol. 1986; 137:702–7. PMID: 3487578
- Bushley KE, Raja R, Jaiswal P, Cumbie JS, Nonogaki M, Boyd AE, et al. The genome of *Tolypocladium inflatum*: evolution, organization, and expression of the cyclosporin biosynthetic gene cluster. PLoS Gen. 2013; 9:e1003496.
- Telleria EL, Martins-da-Silva A, Tempone AJ, Traub-Csekö YM. *Leishmania*, microbiota and sand fly immunity. Parasitol. 2018; 145:1336–1353.
- Jamjoom MB, Ashford RW, Bates PA, Kemp, SJ, Noyes HA. Towards a standard battery of microsatellite markers for the analysis of the *Leishmania donovani* complex. Ann Trop Med Parasitol. 2002; 96:265–270. https://doi.org/10.1179/000349802125000790 PMID: 12061973